Author’s response to reviews

Title: Lessons from a Training Needs Assessment to strengthen the capacity of Routine Immunization Service Providers in Nigeria

Authors:
Linda Arogundade (lindaomosefe@yahoo.com; linda.arogundade@dclnigeria.com)
Titilola Akinwumi (titi.akinwumi@dclnigeria.com; titimakay@gmail.com)
Ebubechi Nwaononiwu (Ebubechi.Nwaononiwu@gmail.com; Ebubechi.Nwaononiwu@dclnigeria.com)
Joshua Ezika (joshykime@gmail.com; joshua.ezika@dclnigeria.com)
Inuwa Yau (inuwabarau@yahoo.com)
Shola Molemodile (sholamole@gmail.com; shola.mole@dclnigeria.com)
Chizoba Wonodi (chizobabw@gmail.com)

Version: 3 Date: 06 Jul 2019

Author’s response to reviews:

Editor Comments and Author Responses

1. Image attribution

-- Please state in the cover letter whether the map depicted in Figure 2 is your own or taken from another source.

Response: The Nigerian map as displayed in Figure 2 isn’t privy to us and we have correctly acknowledged and cited the source where it was obtained and adapted from. Please see Methods section paragraph 3.

2. Author contributions

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response: Done. Please refer to Authors contributions at the tail part of the manuscript which was indeed carried out.
3. Formatting

-- Please provide a 'Declarations' heading.

Response: Done. This has been included and alongside a brief narrative. Please refer to the Declarations section in the tail part of the manuscript as well.

-- Please rename 'Methodology' to 'Methods'.

Response: Done. Please refer to the Methods section.

-- In your Abstract please include your study objective in the 'Background' section.

Response: A study objective has been included in the background sub-section under the Abstract section.

4. Clean manuscript

-- At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Response: Done. Based on this suggestion we have taken conscious steps to remove track changes/highlights, edit and where applicable conduct appropriate formatting, restructuring and improve grammatical range and accuracy which includes tackling missing punctuation and subject-verb agreement problems.